Author’s response to reviews

Title: Eating disorders - knowledge, attitudes, management and experience among Norwegian dentists

Authors:

Ann-Katrin Johansson (Ann-Katrin.Johansson@iko.uib.no)
Anders Johansson (Anders.Johansson@iko.uib.no)
Eva Nohlert (eva.nohlert@ltv.se)
Claes Norring (claes.norring@ki.se)
Anne Nordrehaug Åstrøm (Anne.Aastrom@iko.uib.no)
Åke Tegelberg (ake.tegelberg@orebroll.se)

Version: 2 Date: 14 October 2014

Author’s response to reviews:

Responses to Editorial requests,

Thanks for your comments and please find below a list of changes made to the manuscript according to the editorial requests.

1. Ethics has been addressed in the Method section
2. Competing Interests have been described and inserted after the Abbreviations
3. Authors’ Contributions have been defined and inserted after Competing Interests section
4. A list of abbreviations has been inserted after the Acknowledgement section
5. As regards availability of Supporting Data section, there are no supporting data that we wish to include in this submission.

Bergen, 14 October 2014

Ann-Katrin Johansson